Author’s response to reviews

Title: Factors Predisposing to Shared Governance: A Qualitative Study

Authors:
Foroozan Atashzadeh-Shoorideh (f_atashzadeh@sbmu.ac.ir)
Mohammad Mehdi Sadoughi (sadoughi_mm@sbmu.ac.ir)
Maryam Sattarzadeh Pashabeig (sattarzadehmaryam@yahoo.com)
Alice Khachian (Khachian.a@iums.ac.ir)
Mansoureh Zagheri Tafreshi (m.z.tafreshi@sbmu.ac.ir)

Version: 2 Date: 06 Dec 2018

Author’s response to reviews:

Dear Editor of Journal of BMC Nursing

Thank you for giving us the opportunity to revise our manuscript. We sincerely appreciate the very useful reviewers’ comments and suggestions in our manuscript. Unfortunately, these comments do not relate to our manuscript.

Editor Comments:

BACKGROUND SECTION:
Consider using a balance in your arguments around graduate level preparation e.g. Aiken's work from the USA suggests that mortality is improved with graduate level nurses in surgical units. This section also needs to be edited into paragraphs.

DATA ANALYSIS: Is the word 'Gadget' the proper term? Are you referring to 'medical devices'? Gadget tends to be a lay-person term. Each section within the data codes contain areas of discussion (rather than findings) -suggest that the findings are written and then a separate discussion section.

Response:
I don’t use these works and sentences in our manuscript. I think it related to other manuscript.
Thank you very much for your comments for improving this manuscript. The manuscript has been resubmitted to your journal. We look forward to your positive response.

Regards,

Dr. Foroozan Atashzadeh –Shoorideh,
Associate Professor, Department of Nursing Management, Nursing & Midwifery School, Shahid Beheshti University of Medical Sciences, Tehran, Iran
E-mail: f_atashzadeh@sbmu.ac.ir